Author’s response to reviews

Title: Autonomic nervous system response to remote ischemic conditioning: heart rate variability assessment

Authors:

Daniel Osório (d.osorio@fct.unl.pt)
Ricardo Viana-Soares (ricardo.soares@nms.unl.pt)
João Marto (joao.pedro.seabra.marto@gmail.com)
Marcelo Mendonça (marcelomendoncasousa@gmail.com)
Hugo. Silva (hsilva@plux.info)
Cláudia Quaresma (q.claudia@fct.unl.pt)
Miguel Viana-Baptista (mvianabaptista@nms.unl.pt)
Hugo Gamboa (hgamboa@fct.unl.pt)
Helena L.A. Vieira (helena.vieira@nms.unl.pt)

Version: 2 Date: 07 Aug 2019

Author’s response to reviews:

Dear Editor,

We are herein re-submitting our manuscript BCAR-D-19-00328R1, entitled “Autonomic nervous system response to remote ischemic conditioning: heart rate variability assessment” by Osório et al. We have fulfilled all requests.

In what concerns “Availability of Data and Material” we stated “The datasets used and analysed during the current study are available from the corresponding author on reasonable request”, as it is exemplified in BMC Cardiovascular Disorders webpage.

Looking forward to hearing your opinion on the manuscript.
Yours sincerely,

Helena L. A. Vieira